Contents lists available at ScienceDirect



Journal of Pharmaceutical and Biomedical Analysis

journal homepage: www.elsevier.com/locate/jpba



# Improving the *ex vivo* stability of drug ester compounds in rat and dog serum: Inhibition of the specific esterases and implications on their identity

Matthias Koitka<sup>a,\*</sup>, Joachim Höchel<sup>b</sup>, Hille Gieschen<sup>b</sup>, Hans-Hubert Borchert<sup>c</sup>

<sup>a</sup> INSERM U850, Université de Limoges, Faculté de Médecine, 2 rue du Dr Marcland, 87025 Limoges, France

<sup>b</sup> Bayer Schering Pharma AG, Drug Metabolism and Pharmacokinetics, Muellerstrasse 170-178, 13342 Berlin, Germany

<sup>c</sup> Freie Universität Berlin, Institut für Pharmazie, Kelchstrasse 31, 12169 Berlin, Germany

# ARTICLE INFO

Article history: Received 24 June 2009 Received in revised form 14 September 2009 Accepted 17 September 2009 Available online 23 September 2009

Keywords: Sagopilone Drospirenone Methylprednisolone aceponate Rat and dog serum esterases Mass spectrometry

### ABSTRACT

In drug development, it has been noticed that some drug compounds, especially esters, are unstable in serum samples *ex vivo*. This can lead to a substantial underestimation of the actual drug concentration.

The rat and the dog, representing a rodent and non-rodent species, respectively, are widely used in preclinical studies. We studied the degradation of three structurally different drug esters in rat and dog serum. Moreover, the efficiency of selected enzyme inhibitors to prevent these degradations was investigated. Furthermore, we found indications of the identity of the drug-specific esterases by means of their inhibitor sensitivity as well as by protein purification and identification. The studied drugs were sagopilone, drospirenone, and methylprednisolone aceponate (MPA) all of which are used in (pre-)clinical drug development.

The sagopilone-cleaving esterases in rat serum were inhibited by serine hydrolase inhibitors. We partly purified these esterases resulting in an activity yield of 5% and a purification factor of 472. Using matrix-assisted laser desorption ionization (MALDI)-time of flight (TOF)-mass spectrometry (MS), the rat carboxylesterase isoenzyme ES-1 was identified in these fractions, thus pointing to its involvement in sagopilone cleavage. Drospirenone cleavage in rat serum was effected by butyrylcholinesterase (BChE) and paraoxonase 1 (PON1) as we deduced from the high efficacy of certain serine hydrolase and metallohydrolase inhibitors, respectively. Likewise, some inhibition characteristics implied that MPA was cleaved in rat serum by BChE and serine proteases. Partial purification of the MPA-specific esterases resulted in activity yields of 1–2%, exhibiting up to 10,000-fold purification.

In dog serum, we found that sagopilone was not degraded which was in contrast to MPA and drospirenone. MPA degradation was mainly prevented by serine hydrolase inhibitors. We used a threestep purification to isolate the esterases cleaving MPA. This procedure resulted in an activity yield of 12% and 645-fold purification. By protein identification using liquid chromatography (LC)–electrospray ionization (ESI)–MS, we identified  $\alpha_2$ -macroglobulin ( $\alpha_2$ M) in the active fractions. We therefore assumed that serine hydrolases, probably butyrylcholinesterase, known to form esteratically active complexes with  $\alpha_2$ M, were responsible for MPA cleavage. In contrast, PON1 was assumed to be involved in drospirenone cleavage due to the high efficiency of metallohydrolase inhibitors. This indication was supported by the presence of PON1 in drospirenone-cleaving fractions as we found by affinity chromatography and Western immunoblotting for isolation and detection of PON1, respectively.

The identity of the assumed cleaving enzymes remains, however, to be further studied. The inhibitors we found can serve as a tool for stabilizing drug ester compounds in biological samples *ex vivo*.

© 2009 Elsevier B.V. All rights reserved.

*Abbreviations*:  $\alpha_2 M$ ,  $\alpha_2$ -macroglobulin; AC, affinity chromatography; AChE, acetylcholinesterase; ACN, acetonitrile; albumin-AC, affinity chromatography for removal of albumin; APMSF, 4-amidinophenylmethanesulfonyl fluoride hydrochloride; BCA, bicinchoninic acid; BChE, butyrylcholinesterase; BNPP, bis(*p*-nitrophenyl) phosphate sodium salt; CaCl<sub>2</sub>, calcium dichloride dihydrate; cDNA, complementary deoxyribonucleic acid; CE, carboxylesterase; ChE, cholinesterase; EDTA, ethylenediamine tetraacetic acid disodium salt dihydrate; EGTA, ethyleneglycol-bis(2-aminoethylether)-tetraacetic acid; IEC, anion-exchange chromatography; irinotecan, 7-ethyl-10-[4-(1-piperidino)-1-piperidino)carbonyloxycamptothecin, also referred to as CPT-11; LC-ESI-MS, liquid chromatography-electrospray ionization-mass spectrometry; MALDI-TOF-MS, matrix-assisted laser desorption ionization-time of flight-mass spectrometry; MPA, methylprednisolone aceponate; NaCl, sodium chloride; NAF, sodium fluoride; NCBI, National Center for Biotechnology Information (Bethesda, USA); NPA, 4-nitrophenyl acetate; PAF-AH, platelet-activating factor acetylhydrolase; PCR, polymerase chain reaction; Pefabloc<sup>®</sup> SC, 4-(2-aminoethyl-polyacrylamide gel electrophoresis; SEC, size-exclusion chromatography; TBA, tetrabutylammonium bromide; TBS, Tris buffered saline; TFA, trifluoroacetic acid; TLCK, *N*- $\alpha$ -tosyl-L-lysine chloromethylketone hydrochloride; Tris, tris(hydroxymethyl)aminomethane; Tris-HCl, tris(hydroxymethyl)aminomethane hydrochloride; UV-vis, ultraviolet-visible spectroscopy.

\* Corresponding author. Tel.: +33 555 43 58 95; fax: +33 555 43 59 36.

### 1. Introduction

The stability of drugs in biological samples plays a crucial role in drug development since preclinical and clinical studies require correct drug quantitation with bioanalytical methods. The drug stability mainly depends on the chemical properties of the drug (e.g. the presence of certain functional groups), on the biological properties of the sample matrix (e.g. enzyme activities) as well as on the storage conditions of the drug-containing samples (e.g. temperature conditions). The degradation of drugs after sampling will certainly result in a drug concentration below the true concentration on the time point of sampling. Therefore, authorities responsible for drug marketing approval have published some guidelines for drug development studies in animals and humans in order to examine and to ensure sample stability from sampling to the actual sample analysis [1–4].

The instability of drugs in biological samples was clearly exemplified by drug ester compounds in blood, plasma or serum samples [5,6]. The degradation of drug esters in these matrices was ascribed to esterase activities [7,8]. Esterases belong to the class of enzymes splitting an ester hydrolytically into an acid and an alcohol (hydrolases; EC 3.1) [9,10]. Among others, the following esterases are present in animal and human plasma and serum: acetylcholinesterase (AChE; EC 3.1.1.7), butyrylcholinesterase (BChE; EC 3.1.1.8), carboxylesterase (CE; EC 3.1.1.1), and paraoxonase (PON; EC 3.1.8.1; formerly known as arylesterase) [7,9,11,12]. AChE, BChE and CE are serine hydrolases, i.e. they possess a catalytically active serine residue in their active sites as part of the catalytic triad [9]. In contrast, a catalytic dyad of two histidine residues was found in PON and assumed to be responsible for the esterase and lactonase activities of PON [13]. Furthermore, calcium ions play an important role in the conformational stability and catalytic activity of PON [13], thereby emphasizing the classification of PON as a calcium-dependent metallohydrolase [9]. BChE, CE and PON are considered to be the most relevant esterases in plasma or serum in terms of drug ester cleavage since they exhibit relatively high concentrations in these matrices, on the one hand, and relatively low substrate specificities (as compared to AChE), on the other hand [7,14]. However, there were some drugs which were mainly hydrolyzed by serum albumin [7]. In some cases, the used albumin preparations were contaminated with BChE, thus associating the esterase activity with BChE rather than with albumin [9].

In drug development, one has become aware of the fact that certain drug ester compounds underwent rapid degradation, especially, in rat serum ex vivo. The rat is a typical and widely used rodent species in preclinical studies whereas the dog represents an often employed non-rodent species. When compared to rat serum, the esterase activities of dog and human serum are similar to a great extent. This was demonstrated, on the one hand, quantitatively by incubation studies and, on the other hand, qualitatively by gel electrophoresis studies [15,16]. The esterase activities in rat serum have mostly been reported to be considerably higher than in dog or human serum concerning certain ester compounds [17-22]. In contrast, there are some drug esters which were described to have similar or higher stability in rat serum or plasma compared with the corresponding matrices of dogs or humans [23-26]. Rat serum was found to exhibit a broad CE band near the albumin band in gel electrophoresis [15]. This intensive CE band is, however, not present in dogs and humans [15]. Accordingly, the CE activity of rat serum was reported to be distinctly higher than in dogs or humans, leading to the conclusion that dog and human serum do not contain CE in relevant concentrations [7,27-30]. Rat serum or plasma have, on the other hand, a 3–9-fold lower cholinesterase (ChE) activity than the corresponding dog or human matrix [31,32]. It can thus be assumed that CE substrates may be cleaved faster in rat serum than in dog or human serum. This is in contrast to the hydrolysis of BChE



**Fig. 1.** Structural formula of (A) sagopilone, (B) drospirenone, and (C) methylprednisolone aceponate (MPA). An asterisk marks the position of a <sup>3</sup>H-label.

substrates. For instance, irinotecan (7-ethyl-10-[4-(1-piperidino)-1-piperidino]carbonyloxycamptothecin, also referred to as CPT-11) is a CE substrate which is indeed cleaved more rapidly in rat serum than in human serum [19,33]. Procaine is well known as a BChE substrate which is rapidly degraded in dog and human serum, but stable in rat serum [7,25].

The aim of the present study was to investigate and to improve, where appropriate, the stability of the three structurally different drug esters sagopilone, drospirenone and methylprednisolone aceponate (MPA) in rat and dog serum. Esterase inhibitors were often used successfully for improving the stability of drug ester compounds in animal and human blood, plasma or serum [5,6,34]. In our study, we investigated the efficiency of certain esterase inhibitors to protect the studied drug esters from hydrolysis in rat and dog serum. Furthermore, the different inhibitor sensitivity gave implications on the identity of the responsible esterases. In addition, we isolated the esteratic fractions from native serum and used them for protein identification in order to explain the different inhibitor efficiency towards the drug-cleaving esterases.

In the present study, we investigated the esteratic cleavage of three structurally different drug esters (sagopilone, drospirenone, and methylprednisolone aceponate [MPA]) in rat serum *in vitro*. These drugs have long been or are currently studied in preclinical and clinical drug development, especially at Bayer Schering Pharma AG. Improving their stability in biological matrices is of utmost importance for on-going and future pharmacokinetic studies with these and structurally related compounds. We have chosen these three drugs since they represent different classes of drug esters: Sagopilone (also referred to as ZK-EPO) is a 16-membered *macrolactone* (an epothilone derivative) and presently under clinical investigation as an anti-tumour drug (Fig. 1A) [35]. Drospirenone can be classified as a *steroid lactone* which structurally resembles

spironolactone, both carrying a  $\gamma$ -butyrolactone ring at the spirocarbon atom in position 17 of the steroid structure (Fig. 1B). It is a well-established gestagen used for oral hormonal contraception and postmenopausal hormone replacement therapy [36]. In contrast, MPA is a *corticosteroid with two open-chain ester groups* at position 17 $\alpha$  and 21 (Fig. 1C). It is therapeutically used for topical treatment of various forms of dermal eczema as well as of sunburn, psoriasis and atopical dermatitis [37]. The three investigated compounds were tritium-(<sup>3</sup>H-)labeled for radiochemical HPLC analysis.

In the present study, we found efficient esterase inhibitors which can subsequently be used for stabilization of the three investigated drugs as well as structurally related ester drugs in serum samples during preclinical and clinical studies *ex vivo*.

#### 2. Materials and methods

#### 2.1. Chemicals

All chemicals were of analytical reagent grade unless otherwise stated.

The <sup>3</sup>H-labeled drug ester compounds sagopilone ((1*S*,3*S*,7*S*,10*R*,11*S*,12*S*,16*R*)-7,11-dihydroxy-8,8,12,16-tetra-

methyl-3-(2-methyl-1,3-benzothiazol-5-yl)-10-([2,3-<sup>3</sup>H<sub>2</sub>]prop-2enyl)-4,17-dioxabicyclo[14.1.0]heptadecane-5,9-dione; Fig. 1A), drospirenone  $(6\beta,7\beta,15\beta,16\beta$ -dimethylene-3-oxo-17 $\alpha$ -pregna-4-ene-21,17-[20,21- ${}^{3}H_{2}$ ]carbolactone; Fig. 1B), and MPA  $(21-acetoxy-11\beta-hydroxy-6\alpha-methyl-17-propionyloxy-1,4-[1,2-$ <sup>3</sup>H<sub>2</sub>]pregnadiene-3,20-dione; Fig. 1C) were synthesized by Bayer Schering Pharma AG (Berlin, Germany), 4-Nitrophenyl acetate (NPA: 99%) and tris(hvdroxymethyl)aminomethane hydrochloride (Tris-HCl; 99%) were purchased from Alfa-Aesar (Karlsruhe, Germany). Acetonitrile (ACN; gradient grade), calcium dichloride dihydrate (CaCl<sub>2</sub>), glacial acetic acid (acetic acid, 100%), ethanol (100%), glycerine (87%), water for HPLC (LiChrosolv), methanol (gradient grade), sodium chloride (NaCl) and tris(hydroxymethyl)aminomethane (Tris) were obtained from Merck (Darmstadt, Germany). Ammonia solution (33%) was from Riedel-de Haën (Seelze, Germany). Ammonium hydrogen carbonate (ultrapure) and trifluoroacetic acid (TFA; for protein sequence analysis) were purchased from Sigma-Aldrich (Steinheim, Germany). Water for non-HPLC purposes was produced by a Milli-Q UF Plus ultrafiltration device from Millipore (Eschborn, Germany).

The enzyme inhibitor Pefabloc<sup>®</sup> SC (4-(2-aminoethyl-)benzenesulfonyl fluoride hydrochloride) was purchased from Biomol (Hamburg, Germany). The inhibitors 4-amidinophenylmethanesulfonyl fluoride hydrochloride (APMSF), bis(*p*-nitrophenyl) phosphate sodium salt (BNPP), copper(II) sulfate (99%), ethylenediamine tetraacetic acid disodium salt dihydrate (EDTA; 99%), ethyleneglycol-bis(2-aminoethylether)-tetraacetic acid (EGTA; 98%), *N*- $\alpha$ -tosyl-L-lysine chloromethylketone hydrochloride (TLCK), sodium fluoride (NaF; 99%), tetrabutylammonium bromide (TBA; 99%) and zinc sulfate heptahydrate (99%) were from Sigma–Aldrich (Steinheim, Germany).

# 2.2. Rat serum

Male and female Wistar rats (*Rattus norvegicus*, Crl:WI (Han)), weighing between 155 and 265 g, were delivered by Charles River (Sulzfeld, Germany). They were housed and fed in accordance with institutional and regulatory guidelines. Blood was drawn from the rats by exsanguination *via* the caudal vena cava under anesthesia using isofluran (Forene<sup>®</sup>; Abbott, Wiesbaden, Germany). Serum was pooled from both males and females, equivalently in volume, and stored frozen for not longer than 14 days at approximately -20 °C.

| Table 1   HPLC conditions for the inhibition               | ı studies and drug assays ol                                  | if native serum and th | ie isolated fractions.                                     |                                   |                   |                |  |  |                            |
|--|---|------------------------|--|-----------------------------------|-------------------|----------------|--|--|----------------------------|
|  | Sagopilone  |                        | Drospirenone   |                                   |                   |                | MPA  |  |                            |
| HPLC columns (all from<br>VDS Optilab, Berlin,<br>Germanv) | Kromasil 100 C18, 5 µJ  | .m, 250 × 4.6 mm       | Hypersil ODS 120   | )Å, 3 μm, 125 × 4.6 mm            |                   |                | Spherisorb ODS-2, 3 μո                             | n, 125 × 4.6 mm                                    |                            |
| HPLC eluents   | <ul><li>Eluent 1: acetic acid</li><li>Eluent 2: ACN</li></ul> | 0.1 % (v/v)            | <ul><li>Eluent 3: acetic</li><li>Eluent 4: metha</li></ul> | acid 0.1 % (v/v), pH 7.5,<br>anol | adjusted with amm | nonia solution | • Eluent 5: ACN/water 3<br>• Eluent 6: ACN/water 7 | 30%/70% + 0.15% (v/v/v)<br>70%/30% + 0.15% (v/v/v) | acetic acid<br>acetic acid |
|  | (min)   | 1                      | 2  | (min)                             | ŝ                 | 4              | (min)  | 5  | 9                          |
| HPLC gradient of the                                       | 0   | 80                     | 20   | 0                                 | 75                | 25             | 0  | 100  | 0                          |
| eluents 1 and 2, 3 and                                     | 8   | 40                     | 60   | 35                                | 45                | 55             | 20   | 0  | 100                        |
| 4, or 5 and 6 (%, v/v)                                     | 20  | 40                     | 60   | 55                                | 0                 | 100            | 25   | 0  | 100                        |
| over run time (min)  | 23  | 20                     | 80   | 58                                | 0                 | 100            | 35   | 100  | 0                          |
|  | 25  | 20                     | 80   | 60                                | 75                | 25             |  |  |                            |
|  | 27  | 80                     | 20   |                                   |                   |                |  |  |                            |

#### 2.3. Dog serum

Male and female Beagle dogs (*Canis familiaris*), weighing between 6.2 and 9.0 kg, were delivered by Marshall BioResources (North Rose, USA). They were housed and fed in accordance with institutional and regulatory guidelines. Blood was drawn from the jugular vein and processed to serum. Serum was pooled, equivalently in volume, from both males and females. It was used immediately or stored for not longer than one day at approximately  $4 \degree C$ .

# 2.4. Inhibition studies of native serum

Initial incubation experiments in rat and dog serum have shown that drospirenone and MPA were both degraded in rat and dog serum whereas sagopilone was cleaved in rat serum only. To find efficient enzyme inhibitors, selected esterase and protease inhibitors (see Section 2.1) were tested for their potency to prevent degradation of sagopilone, drospirenone, and MPA in rat serum as well as of MPA and drospirenone in dog serum. The selection of enzyme inhibitors was mainly based on their reported solubility and stability in water (or buffered solutions), their inhibitory properties (e.g. irreversible inhibitors were preferred over reversible ones), their toxicity, and their inexpensive availability since these inhibitors would be used in serum (or even blood) samples during preclinical and clinical studies.

All the chosen inhibitors were dissolved in water, apart from EGTA which was prepared in a Tris-HCl-buffered solution (100 mM) containing 10 mM CaCl<sub>2</sub> (pH 7.4), due to the higher solubility of EGTA in this buffer. Ninety microlitre of native serum were pre-incubated with 10 µL of the inhibitor solution for 30 min in a water bath at  $37 \pm 0.5$  °C. This mixture was then given to one of three drugs (whose solutions in ethanol had been dried before under a gentle stream of nitrogen) and incubated for 2 min, 6 h or 20 min (referring to sagopilone, drospirenone or MPA in rat serum, respectively) or for 6h (referring to drospirenone or MPA in dog serum) at  $37 \pm 0.5$  °C. The final concentrations (and specific radioactivity) of the drugs were as follows:  $0.15 \,\mu\text{M}$  (580 GBq/mmol) for sagopilone, 10 µM (17 GBq/mmol) for drospirenone, and 0.2 µM (832 GBq/mmol) for MPA in rat serum, respectively, and 0.2 µM (832 GBq/mmol) or 1 µM (83 GBq/mmol) for MPA, and 10 µM (17 GBq/mmol) for drospirenone in dog serum, respectively. The final concentrations of the inhibitors are given in Tables 2 and 3. Corresponding positive controls were prepared using 10 µL of a Tris-HCl-buffered solution (100 mM), containing 10 mM CaCl<sub>2</sub> (pH 7.4), instead of the inhibitor solution. The reaction in each sample was stopped by adding 100 µL ACN and vortexing. After centrifugation, the supernatants were dried under a gentle stream of nitrogen. The residues were dissolved in 100  $\mu$ L of eluent 1, 3 or 5 (referring to sagopilone, drospirenone or MPA, respectively; see Table 1) and stored at 10 °C for subsequent HPLC analysis. Twenty microlitre of each sample were injected onto the column. Table 1 summarizes the drug-specific chromatography conditions. The HPLC system consisted of a P680A HPG-2 pump (set at 1 mL/min), an ASI-100T autosampler (set at 10 °C), a TCC-100 MSV column oven (set at 22 °C), an UVD 170U ultraviolet-visible spectroscopy (UV-vis) detector (all from Dionex Softon, Germering, Germany), and a radiochemical detector (Radiomatic 525TR; Packard Instrument, Meriden, USA) using Ultima-Flo M (PerkinElmer, Shelton, USA) as liquid scintillation cocktail. The chromatograms were evaluated by use of Chromeleon 6.50 software (Dionex Softron, Germering, Germany). The peak identity of the drugs and their cleavage products was assessed in former internal studies by co-chromatography with the reference compounds. The quantitation of the drugs and their corresponding cleavage products was accomplished by determining the ratio of the peak area of the parent compound to the cleavage product(s) as measured by radiochemical detection. One sample per drug and inhibitor concentration was analyzed. The ratio of the relative peak area of the cleaved drug in the inhibited sample to that of the cleaved drug in the positive control sample was calculated. The difference between this ratio and 100% was defined as inhibition (in %).

### 2.5. Esterase activity assay using NPA

General esterase activity of the isolated fractions from the purification steps (see Section 2.8 and Fig. 2) was determined by a kinetic assay using the non-specific esterase substrate NPA which we have already used in our previous study [38]. Two hundred microlitres of each fraction were mixed with 10 µL of a NPA solution (50 mM; in methanol) in wells of a microplate (Nunc F96; Nunc, Wiesbaden, Germany). Then, the change of the absorbance was measured every 15 s over 3 min at a wavelength of 405 nm in a spectrophotometer (SpectraMax 190; Molecular Devices, Sunnyvale, USA) set at 30 °C. The measurement was evaluated by SoftMax Pro software (version 4.7; Molecular Devices) giving the maximum slope of a linear regression curve from the absorbance values between the fixed time points. Blank samples (using purification buffers only instead of fractions) were measured. Fractions with higher maximum slopes than those of the respective blank samples were further considered for PAGE analyses (see Section 2.10), protein determination (see Section 2.9), drug assays (see Section 2.6), and the next step of the purification sequence (see Section 2.8).

# 2.6. Esterase activity assay using sagopilone, drospirenone, or MPA

The fractions from the purification steps (see Section 2.8) were washed with a Tris-HCl-buffered solution (100 mM, pH 7.4) containing 10 mM CaCl<sub>2</sub>. Then, the washed fractions were concentrated by using Amicon Ultra-15 ultrafiltration device (MW cut-off: 3 kDa; Millipore, Cork, Ireland). One hundred microlitre of the concentrated fractions, or native serum, were added to the given drug (whose solution in ethanol had been dried under a gentle stream of nitrogen) and incubated in a water bath at  $37 \pm 0.5$  °C. The final concentrations (and specific radioactivity) of the drugs in the incubation mixtures were as follows: 3 µM (55 GBq/mmol) for sagopilone,  $10 \,\mu\text{M}$  (17 GBq/mmol) for drospirenone, and  $0.2 \,\mu\text{M}$ (832 GBq/mmol) for MPA in rat or dog serum, respectively. The incubation procedure and HPLC analyses were done as described for the inhibition studies (see Section 2.4). The time periods for incubation in native, or fractions from, rat serum were 0, 1, 5, 30, and 60 min for sagopilone (serum), 0, 5, and 60 min for sagopilone (fractions), 0, 10, 30, 60, and 120 min for drospirenone (serum), 0, 1, and 4h for drospirenone (fractions), 0, 1, 5, 10, and 20 min for MPA (serum), and 0, 5, and 10 min for MPA (fractions). The time periods for incubation in native, or fractions from, dog serum were 0, 1, 2, 4, and 6 h for sagopilone (serum), 0, 1, 3, 6, and 12 h for drospirenone (serum), 0, 6, and 10 h for drospirenone (fractions), 0, 1, 4, 8, and 12 h for MPA (serum), and 0, 6, and 10 h for MPA (fractions). One sample per drug, incubation time point and pooled fraction, or serum type, was analyzed. Using the relative peak areas (which were calculated as described for the inhibition studies), the amount of the parent compound (in pmol) was calculated with the help of the known initial drug concentration in the incubation mixture. The amounts of the parent compounds were correlated with the incubation time. The activity  $(in \mu U)$  was defined as the greatest slope of the linear regression curves between the fixed time points in accordance with the data evaluation of the NPA assay (see Section 2.5). One unit (U) of activity represents the amount of enzyme hydrolyzing 1 µmol of drug in 1 min under assay conditions.



Fig. 2. Flow chart of the purification procedure of esteratic fractions from rat and dog serum with keys to applied esterase activity assays and other analytical techniques.

# 2.7. Buffers used for the purification steps

The buffers A, B, and C were used for the anion-exchange chromatography step (see Section 2.8). Buffer A consisted of a Tris–HCl-buffered solution (20 mM) containing 10 mM NaCl and 10% (v/v) glycerine (pH 7.6). Buffer B was a Tris–HCl-buffered solution (20 mM) containing 200 mM NaCl and 10% (v/v) glycerine (pH 7.6). Buffer C was a Tris–HCl-buffered solution (20 mM) containing 500 mM NaCl and 10% (v/v) glycerine (pH 7.6).

Buffer D, used for size-exclusion chromatography (see Section 2.8), consisted of a Tris–HCl-buffered solution (20 mM) containing 150 mM NaCl and 10% (v/v) glycerine (pH 7.6).

The buffers E and F were used for affinity chromatography removing albumin (see 2.8). Buffer E was a Tris–HCl-buffered solution (20 mM) containing 10% (v/v) glycerine (pH 7.0). Buffer F consisted of a Tris–HCl-buffered solution (20 mM) containing 1.5 M NaCl and 10% (v/v) glycerine (pH 7.0).

The buffers G, H, and I were prepared for the affinity chromatography isolating PON1-containing fractions (PON-AC, see Section 2.8). Buffer G was a Tris–HCl-buffered solution (20 mM) containing 2 M NaCl, 1 mM CaCl<sub>2</sub>, 5  $\mu$ M EDTA, and 10% (v/v) glycerine (pH 8.0). Buffer H consisted of a Tris–HCl-buffered solution (20 mM) containing 1 mM CaCl<sub>2</sub>, 5  $\mu$ M EDTA, and 10% (v/v) glycerine (pH 8.0). Buffer I was a Tris–HCl-buffered solution (20 mM) containing 0.1% (w/v) sodium deoxycholate, 1 mM CaCl<sub>2</sub>, 5  $\mu$ M EDTA, and 10% (v/v) glycerine (pH 8.0).

All the buffers were filtered and degassed prior to use.

# 2.8. Isolation and purification of esteratic fractions

The isolation and purification scheme comprised three sequential steps (methods I–III, see below and Fig. 2) which were mainly orientated on the method of Tsuji et al. [19] with some alterations [39]. In addition, PON1-containing fractions from dog serum were obtained by method IV (i.e. PON-AC; see below) essentially based on the procedure of Gan et al. [40]. All chromatographic steps were carried out on the protein purification system ÄKTAprime plus (GE Healthcare, Uppsala, Sweden) at approximately 4°C, except from the methods III and IV which were accomplished at ambient temperature. The runs were monitored by an integrated UV detector measuring at 280 nm.

### 2.8.1. Anion-exchange chromatography (IEC)

Frozen, pooled rat serum was thawed at ambient temperature. An aliquot was assayed for total protein content (see Section 2.9) and esterase activities with regard to the three drugs (see Section 2.6). Six millilitre of serum were diluted with buffer A to 75 mL. An anion-exchange chromatography column (DEAE Sephacel matrix filled in an XK 26/20 column, GE Healthcare, Uppsala, Sweden) was loaded with the prepared serum. Throughout the run, a constant flow rate of 2 mL/min was used. After washing with buffer A, the proteins eluted from the column during a linear gradient from 10 to 200 mM NaCl (buffer A and B, respectively). Then, the column was washed by a one-step gradient from 200 to 500 mM NaCl (buffer B and C, respectively). Finally, the starting conditions were restored (buffer A). Fractions (10 mL each) were collected during the run. The fractions with sufficient esterase activity (as measured by the NPA assay; see Section 2.5) were further used for PAGE analyses and protein determination. Based on PAGE results, qualitatively similar fractions were pooled for the drug assays. Pooled fractions with the highest drug-specific activity yields and purification factors were prepared for the next purification step (see Fig. 2).

#### 2.8.2. Size-exclusion chromatography (SEC)

The active fractions from the IEC were washed with buffer D and concentrated. An XK 26/70 column was packed with Sephacryl S-300 HR (GE Healthcare, Uppsala, Sweden). After sample application, elution was carried out at 0.5 mL/min with approximately 300 mL of buffer D, collecting fractions of 5 mL volume. The fractions were analyzed in the same way as after the IEC step. Those with the highest drug-specific activities were chosen to be prepared for the next purification step.

For determination of molecular mass, the following standard proteins were used to calibrate the column (Gel Filtration Standard; Bio-Rad Laboratories, Hercules, USA): bovine thyroglobulin (670 kDa), bovine  $\gamma$ -globulin (158 kDa), chicken ovalbumin (44 kDa), horse myoglobin (17 kDa), and vitamin B<sub>12</sub> (1.35 kDa).

# 2.8.3. Affinity chromatography (AC) for removal of albumin (albumin-AC)

The pooled fractions from the SEC were washed with buffer E by ultrafiltration and concentrated. An affinity chromatography column filled with 5 mL of a Cibacron Blue F3G-A-coupled matrix (HiTrap Blue HP; GE Healthcare, Uppsala, Sweden) was employed for removal of albumin. Following sample application, the column was washed with the NaCl-free buffer E. Then, the bound proteins were eluted by a one-step gradient from 0 to 1.5 M NaCl (buffer E and F, respectively). Eventually, the starting conditions were restored (buffer E). Fractions (2 mL each) were collected and taken for analysis in the same way as after the IEC step. The fractions from rat or dog serum exhibiting the highest activity for sagopilone or MPA cleavage were subjected to protein identification (see Section 2.11 or Section 2.12), respectively.

Applying the purification sequence on dog serum esterases resulted in a total activity loss of the SEC fractions and albumin-AC fractions as determined by the NPA assay (see Section 2.5). This was in marked contrast to the IEC fractions from dog serum as well as to all of the fractions from rat serum. Therefore, all the SEC fractions as well as albumin-AC fractions from dog serum with UV absorbance values greater than that of buffer D as well as buffer E or F, respectively, were further used for PAGE analyses (see Section 2.10), protein determination (see Section 2.9) and drug assays (see Section 2.6).

# 2.8.4. Affinity chromatography (AC) for isolation of PON1-containing fractions (PON-AC) from dog serum

To isolate PON1-containing fractions from dog serum, a dyeligand affinity chromatography was carried out using a 5-mL column filled with a Cibacron Blue F3G-A-coupled matrix (HiTrap Blue HP; GE Healthcare, Uppsala, Sweden). One millilitre of dog serum was diluted with buffer G to 2 mL and applied to the column. After washing, elution of PON1-containing fractions was carried out by a one-step gradient from 2 to 0 M NaCl (buffer G and H, respectively). The remaining PON1-containing fractions were eluted off the column by changing from buffer H to buffer I. Then, the initial conditions were restored with buffer G. Fractions (2 mL each) were collected and assayed for protein concentration (see Section 2.9). In addition, all the fractions which made part of a peak in the UV chromatogram were taken for sodium dodecyl sulfate-(SDS-)PAGE analysis with Western blot detection (see Section 2.13).

# 2.9. Protein determination

Protein concentration was determined with the bicinchoninic acid (BCA) assay [41]. Referring to the IEC fractions,  $10 \,\mu$ L were mixed with 200  $\mu$ L of the BCA working solution (BCA Protein Assay Kit; Thermo Scientific, Rockford, USA) and incubated for 30 min at 37 °C. As the SEC, albumin-AC, and PON-AC fractions usually contained less than 200  $\mu$ g/mL protein, the micro-BCA assay was employed [42]. Therefore, 100  $\mu$ L were mixed with 100  $\mu$ L of the micro-BCA working solution (Micro BCA Protein Assay Kit; Thermo Scientific) and incubated for 2 h at 37 °C. In both procedures, the standard curve was prepared from serial dilutions of bovine serum albumin.

# 2.10. Sodium dodecyl sulfate-(SDS-)PAGE and native (non-denaturating) PAGE

Two different PAGE techniques, i.e. SDS-PAGE and native PAGE, were used to reveal the purity of the fractions from the purification steps. It is known that both techniques differ in their separation parameters, thereby providing an improved distinction of qualitatively unequal fractions [43].

For SDS-PAGE analysis, NuPAGE 4–12% Bis–Tris gels were chosen (Invitrogen, Carlsbad, USA). Prior to loading, the samples were incubated with a buffer containing 2% (w/v) lithium dodecyl sulfate and 10% (v/v) 2-mercaptoethanol for 10 min at 70 °C. The gels were run in a buffer containing 50 mM 3-(*N*-morpholino)propane sulfonic acid and 0.1% (m/v) SDS at a constant voltage of 200 V for approximately 45 min. After the run, the proteins were stained by use of Coomassie Blue following the manufacturer's instructions (Colloidal Blue Stain Kit; Invitrogen).

Regarding native PAGE analysis, Novex 8–16% Tris–Glycine gels were employed (Invitrogen). The samples were mixed with a buffer containing 50 mM Tris–HCl and loaded onto the gels. The electrophoresis was carried out in a buffer containing 2.5 mM Tris and 19.2 mM glycine at a constant voltage of 125 V for approximately 90 min. Detection was carried out as mentioned before.

# 2.11. Protein identification by matrix-assisted laser desorption ionization (MALDI)-time of flight (TOF)-mass spectrometry (MS)

For protein identification, the 70-kDa protein band of a Coomassie-stained SDS-PAGE gel (from a sagopilone-cleaving albumin-AC fraction purified from rat serum) was cut out, minced and treated with a mixture (50%/50%; v/v) of ACN and a solution of 50 mM ammonium hydrogen carbonate. After decolorization and bleaching, the material was incubated with a solution containing 50 mM ammonium hydrogen carbonate and  $10 \text{ ng/}\mu\text{L}$  trypsin (Sequencing Grade Modified Trypsin; Promega, Mannheim, Germany) overnight at 37 °C. The tryptic digestion was stopped by adding 2% (v/v) TFA. The gel pieces were extracted using a mixture of water/ACN/TFA (39.9%/60%/0.1%; v/v/v) The supernatants were dried under vacuum, the residues then re-dissolved in 0.1% (v/v) TFA and used for MALDI-TOF-MS analysis with a matrix containing  $\alpha$ -cyano-4-hydroxycinnamic acid ( $\alpha$ -CHCA Matrix Kit; LaserBio Labs, Sophia-Antipolis, France). The analysis was carried out on 4700 Proteomics Analyzer (Applied Biosystems, Foster City, USA) and evaluated by Mascot software (Matrix Science, London, UK). The peptides were identified by matching the resulting MS and MS/MS spectra to reported proteins in the NCBI database (National Center for Biotechnology Information, Bethesda, USA). To estimate the probability of a false-positive match, "protein scores" of greater than 67 were considered significant for an actual match (p < 0.05) [44].

# 2.12. Liquid chromatography (LC)–electrospray ionization (ESI)–mass spectrometry (MS)

For protein identification, the 140-kDa protein band of the SDS-PAGE gel (from an MPA-cleaving albumin-AC fraction purified from dog serum) was cut out and minced. The gel pieces were incubated with a mixture (50%/50%; v/v) of acetonitrile (ACN) and a solution of 50 mM ammonium hydrogen carbonate. After centrifugation, decolorization and bleaching, the material was incubated with a solution containing 50 mM ammonium hydrogen carbonate and 10 ng/µL trypsin (Sequencing Grade Modified Trypsin; Promega, Mannheim, Germany) at 37 °C. The tryptic digestion was stopped by adding 2% (v/v) trifluoroacetic acid (TFA). After extraction with a mixture of water/ACN/TFA (39.9%/60%/0.1%; v/v/v), the supernatants were united, dried, and re-dissolved in 0.1% (v/v) TFA. Ten microlitre were injected into the 4000 Q TRAP LC/MS/MS system (linear ion trap) (Applied Biosystems, Foster City, USA). The LC was carried out on the Eksigent NanoLC-1Dplus system (Eksigent Technologies, Dublin, USA) using an Acclaim PepMap column (stationary phase: C18; 15 cm, 75 µm i.d., particle size: 5 µm, pore size: 300 Å) (Dionex, Sunnyvale, USA) with a C18-Trap column (Agilent, Waldbronn, Germany). Two eluents were prepared: (A) 0.1% (v/v) formic acid and (B) ACN / water / formic acid (95%/4.9%/0.1%; v/v/v). A typical LC gradient was run as follows (in % [v/v]): 0–5 min: A/B 95/5; 5–30 min: A/B from 95/5 to 60/40; flushing the column with A/B >60/<40; restoration of the initial conditions (A/B 95/5). Data evaluation was carried out using Mascot software (Matrix Science, London, UK). Matching of the peptides and estimation of the probability of a false-positive match was done as described before (see Section 2.11).

# 2.13. Western blot

To directly detect PON1 in the PON-AC fractions from dog serum, SDS-PAGE analysis of these fractions as well as of dog serum was carried out as described in Section 2.10, except from

#### Table 2

Inhibition of drug cleavage in native rat serum by different enzyme inhibitors. Inhibition (in %) was defined as the difference between 100% and the ratio of the relative peak area of the hydrolyzed drug in the inhibited sample to that of the hydrolyzed drug in the positive control (which contained buffer instead of inhibitor solution). For experimental conditions, see 2.4.

| Inhibitor (concentration)        | Inhibition (%) |                   |      |
|----------------------------------|----------------|-------------------|------|
|                                  | Sagopilone     | Drospirenone      | MPA  |
| BNPP (1 mM)                      | 97.4           | 64.0              | 90.7 |
| BNPP (5 mM)                      | 95.4           | 55.0              | 94.8 |
| EDTA (20 mM)                     | 26.5           | 67.6              | 2.1  |
| NaF (10 mM)                      | 74.2           | n.d. <sup>a</sup> | n.d. |
| NaF (100 mM)                     | 92.7           | 50.6              | 56.5 |
| Pefabloc <sup>®</sup> SC (1 mM)  | 92.1           | 61.8              | 92.3 |
| Pefabloc <sup>®</sup> SC (10 mM) | 89.8           | 66.1              | 93.2 |
| TBA (10 mM)                      | 87.1           | n.d.              | n.d. |
| TBA (100 mM)                     | 95.8           | 99.4              | 22.2 |
| TLCK (10 mM)                     | 82.9           | 19.1              | n.d. |
| Zinc sulfate (10 mM)             | 19.8           | 0.0               | 54.9 |

<sup>a</sup> n.d.: not determined.

protein staining. After electrophoresis, the gel was blotted onto a nitrocellulose membrane (iBlot Gel Transfer Stacks, Regular; Invitrogen, Carlsbad, USA) according to the manufacturer's instructions [45]. Then, the membrane was incubated with blocking buffer (i.e. a solution of 0.05 M Tris-buffered saline (TBS) containing 0.05% (w/v) Tween 20 and 5% (w/v) bovine serum albumin (pH 8.0) (all ingredients from Sigma-Aldrich, Steinheim, Germany). The primary antibody (anti-paraoxonase 1 antibody, produced in rabbit; #P0123, Sigma-Aldrich, Steinheim, Germany) was diluted 1:1000 (v/v) in blocking buffer and added to the membrane. After washing, the secondary antibody (anti-rabbit IgG [whole molecule], alkaline phosphatase conjugate, developed in goat; #A7778, Sigma-Aldrich, Steinheim, Germany) was diluted 1:3000 (v/v) in blocking buffer and incubated with the membrane. Following washing, a solution of nitro blue tetrazolium and 5-bromo-4-chloro-3-indolyl phosphate (Premixed BCIP/NBT Solution; Sigma-Aldrich, Steinheim, Germany) was added to visualize PON1-containing protein bands. Negative control samples using solutions of AChE (from Electrophorus electricus [electric eel]), BChE (from horse serum), porcine liver esterase (all from Sigma-Aldrich, Steinheim, Germany), and protein-free SDS-PAGE sample buffer were analyzed either.

# 2.14. Evaluation of kinetic data

All data from the inhibition studies (see Section 2.4) as well as from the drug assays (see Section 2.6) were computed and evaluated using SigmaPlot software (version 10.0) in connection with SigmaStat (version 3.5) and Enzyme Kinetics Modul (version 1.3) (all from Systat Software, Richmond, USA).

### 3. Results

### 3.1. Inhibition studies of native rat serum

The serine hydrolase inhibitors BNPP and Pefabloc<sup>®</sup> SC were highly effective inhibitors of sagopilone and MPA cleavage in rat serum (>90% inhibition; Table 2). Drospirenone degradation was less influenced by these inhibitors. In contrast, the metallohydrolase inhibitor EDTA had a greater effect on drospirenone than on sagopilone or MPA cleavage. NaF was a more efficient inhibitor of sagopilone cleavage compared to the cleavage of the other drugs whereas TBA equally inhibited sagopilone and drospirenone hydrolysis. TLCK sufficiently prevented sagopilone but not drospirenone cleavage. Zinc sulfate was a weak inhibitor of MPA hydrolysis, showing virtually no effect on the cleavage of the two other drugs.

#### Table 3

Inhibition of drug cleavage in native dog serum by different enzyme inhibitors. For definition of inhibition (in %) see Table 2; for experimental conditions, see Section 2.4.

| Inhibitor (concentration) | Inhibition (%) |      |
|---------------------------|----------------|------|
|                           | Drospirenone   | MPA  |
| APMSF (10 mM)             | 58.7           | 75.8 |
| BNPP (5 mM)               | 31.2           | 70.2 |
| EDTA (20 mM)              | 95.6           | 37.2 |
| EGTA (10 mM)              | 95.4           | 30.4 |
| Copper(II) sulfate (1 mM) | 73.6           | 30.0 |
| NaF (100 mM)              | 94.2           | 9.6  |
| Pefabloc® SC (10 mM)      | 78.9           | 82.3 |
| TBA (100 mM)              | 93.9           | 32.8 |
| TLCK (10 mM)              | 75.0           | 88.1 |
| Zinc sulfate (1 mM)       | 69.4           | 2.5  |

### 3.2. Inhibition studies of native dog serum

Referring to MPA cleavage in dog serum, the serine hydrolase inhibitors Pefabloc<sup>®</sup> SC, APMSF, BNPP, and TLCK were found to be highly efficient inhibitors (70–88% inhibition; Table 3) whereas all the other tested inhibitors exhibited no relevant inhibition. With regard to drospirenone stabilization, the metallohydrolase inhibitors EDTA and EGTA were as efficient as the enzyme inhibitors NaF and TBA (94–95% inhibition each; Table 3). Pefabloc<sup>®</sup> SC (exhibiting 79% inhibition) was more efficient than APMSF (59% inhibition) or BNPP (no marked inhibition) in preventing drospirenone cleavage. The PON inhibitors copper sulfate and zinc sulfate as well as TLCK revealed high inhibition (69–75%) concerning drospirenone hydrolysis.

# 3.3. Purification of esteratic fractions from rat serum and protein identification by MALDI-TOF-MS

The isolation and purification of esteratic fractions from rat serum by use of a sequential, three-step purification protocol was successful in enriching the drug-specific esterase activities. This is demonstrated by high purification factors ranging from 83 to approximately 10,000 (Table 4). Especially, the last purification step (i.e. an affinity chromatography for removal of albumin, referred to as albumin-AC) greatly increased the specific activities of the fractions. The first purification step (i.e. an anion-exchange chromatography, referred to as IEC), however, resulted in a great loss of esteratic activities giving low activity yields in the end of the purification process. Referring to sagopilone, the pooled fractions with the highest activity from the first purification step eluted on the descending part of a peak maximum (corresponding to approximately 70 kDa or less) in the size-exclusion chromatography (referred to as SEC; chromatograms not shown). The activity for sagopilone cleavage was clearly separated from the proteins which were bound to the Cibacron Blue F3G-A-coupled matrix (chromatogram not shown). Consequently, the esterases cleaving sagopilone did not bind to this dye-ligand affinity column. The bound proteins mainly consisted of serum albumin as revealed by SDS-PAGE (results not shown). With respect to drospirenone and MPA, the chromatograms and their activity distributions were comparable to those of sagopilone (data not shown). Thus, the major esteratic activities concerning the three drugs were not associated with serum albumin.

The SDS-PAGE of the albumin-AC fractions cleaving the three drugs unveiled several protein bands (Fig. 3), indicating that the fractions were only partially purified. The Coomassie-stained protein bands of the SDS-PAGE gel of the active fractions hydrolyzing sagopilone were investigated by protein identification using tryptic digestion and MALDI-TOF-MS. At approximately 70 kDa (Fig. 3A),

we were able to match the individual MS/MS spectra of two peptides (GNWAHLDQLAALR and LGIWGLFSTGDEHSR) of this protein band to the rat CE isoenzyme ES-1 (NCBI no. gi|2506388). The protein score of 144 for this match was significant (p < 0.05). We found, however, that rat serum albumin (NCBI no. gi|12402861) was also present in this protein band. Although not investigated, the finding that a protein band at approximately 68 or 70 kDa occurred in the SDS-PAGE gels of the drospirenone- or MPA-cleaving fractions, respectively, suggested the existence of ES-1 and albumin in these fractions either (Fig. 3B and C).

# 3.4. Purification of esteratic fractions from dog serum and protein identification by LC-ESI-MS

Sagopilone was not degraded in dog serum within the incubation time (results not shown). We therefore only investigated the esterases cleaving MPA and drospirenone by protein isolation and identification techniques.

The three-step purification protocol used for isolation of the esterases cleaving MPA resulted in an activity yield of approximately 12% with 645-fold purification (Table 5). The pooled fractions with the highest activity from the first purification step (IEC) eluted at the first, small peak maximum of the SEC corresponding to molecular masses of above 484 kDa (chromatograms not shown). The esterases were firmly bound to the Cibacron Blue affinity column (used for albumin-AC) and only recovered using a high-salt buffer (chromatogram not shown). With the help of the albumin-AC, the specific activity of the MPA-cleaving fractions was strongly enhanced as compared to both preceding purification steps (Table 5).

This purification scheme was not successful referring to drospirenone-cleaving esterases since the found esteratic activity of the active IEC fractions was completely lost after the SEC step (results not shown). PON1 was thus assumed to be involved in drospirenone cleavage. The PON-AC yielded three groups of PON1containing fractions which sequentially eluted from the column by washing with a NaCl-free buffer, elution with sodium deoxycholate, and, finally, restoration of the starting conditions with a high-salt buffer (chromatogram not shown). The main PON1 pro-



**Fig. 3.** SDS-PAGE gels of the albumin-AC fractions from rat serum cleaving sagopilone (A), drospirenone (B), and MPA (C). Lane A, B, and C contained approximately 4.9, 8.1, and 0.2 μg protein, respectively. Detection was accomplished by Coomassie Blue staining (see Section 2.10). The lane designated by "(St.)" shows the bands of the following molecular-mass standard proteins (See Blue Plus2 Pre-Stained Standard; Invitrogen, Carlsbad, USA): myosin (191 kDa), phosphorylase (97 kDa), bovine serum albumin (64 kDa), glutamic dehydrogenase (51 kDa), alcohol dehydrogenase (39 kDa), carbonic anhydrase (28 kDa), myoglobin red (19 kDa), and lysozyme (14 kDa). Two other standard proteins (aprotinin and insulin, B chain) exhibited bands below 14 kDa but were only poorly resolved. The arrows in gel A, B, and C indicate protein bands at 70, 68, and 70 kDa, respectively. The 70-kDa band in gel A was excised and subjected to tryptic digestion and, subsequently, MALDI-TOF-MS (see Section 2.11). Thereby, the rat carboxylesterase isoenzyme ES-1 (contaminated with rat serum albumin) was found.

# Table 4

Summary of purification factors and activity yields of sagopilone-, drospirenone-, and methylprednisolone aceponate- (MPA-)cleaving fractions from rat serum. One unit (U) of activity is defined as the amount of enzyme cleaving 1 µmol of drug in 1 min under assay conditions (see Section 2.6). Specific activity was calculated by dividing the total activity by the total protein content of the fractions from the distinct purification steps. Purification factors were obtained by the ratio of the specific activity of the fractions to the specific activity of rat serum. Activity yields give the ratio of the total activity of the fractions to that of serum.

|              | Purification step | Fraction volume (mL) | Total activity ( $\mu U$ ) | Total protein (mg) | Specific activity<br>(µU/mg) | Purification<br>factor (fold) | Activity yield (%) |
|--------------|-------------------|----------------------|----------------------------|--------------------|------------------------------|-------------------------------|--------------------|
| Sagopilone   | Rat serum         | 6                    | 255                        | 510                | 0.50                         | 1.0                           | 100.0              |
|              | IEC               | 30                   | 30.5                       | 44.6               | 0.69                         | 1.4                           | 12.0               |
|              | SEC               | 15                   | 43.1                       | 18.5               | 2.33                         | 4.7                           | 16.9               |
|              | Albumin-AC        | 4                    | 11.4                       | 0.05               | 236                          | 472                           | 4.5                |
| Drospirenone | Rat serum         | 6                    | 236                        | 510                | 0.52                         | 1.0                           | 100.0              |
|              | IEC               | 30                   | 12.5                       | 42.3               | 0.30                         | 0.6                           | 5.2                |
|              | SEC               | 25                   | 12.8                       | 15.9               | 0.80                         | 1.8                           | 5.4                |
|              | Albumin-AC        | 4                    | 3.4                        | 0.08               | 42.9                         | 82.5                          | 1.4                |
| MPA          | Rat serum         | 6                    | 49.6                       | 501                | 0.10                         | 1.0                           | 100.0              |
|              | IEC               | 20                   | 14.7                       | 6.74               | 2.18                         | 21.8                          | 29.6               |
|              | SEC               | 15                   | 5.0                        | 0.56               | 9.00                         | 90.0                          | 10.1               |
|              | Albumin-AC        | 4                    | 1.2                        | 1.24 µg            | 996                          | 9956                          | 2.4                |

#### Table 5

Summary of purification factors and activity yields of drospirenone- and methylprednisolone aceponate- (MPA-)cleaving fractions from dog serum. For definition of one unit (U) of activity, specific activity, purification factors, and activity yields, see Table 4.

|              | Purification step | Fraction volume (mL) | Total activity ( $\mu U$ ) | Total protein (mg) | Specific activity<br>(µU/µg) | Purification<br>factor (fold) | Activity yield (%) |
|--------------|-------------------|----------------------|----------------------------|--------------------|------------------------------|-------------------------------|--------------------|
| MPA          | Dog serum         | 6                    | 0.87                       | 331                | 2.63                         | 1.0                           | 100.0              |
|              | IEC               | 50                   | 0.22                       | 19.6               | 11.4                         | 4.3                           | 25.6               |
|              | SEC               | 35                   | 0.13                       | 0.68               | 192                          | 73.1                          | 14.9               |
|              | Albumin-AC        | 22                   | 0.10                       | 0.06               | 1697                         | 645                           | 11.8               |
| MPA          | Dog serum         | 1                    | 0.15                       | 59.6               | 2.44                         | 1.0                           | 100.0              |
|              | PON-AC            | 114                  | 0.03                       | 1.91               | 16.7                         | 6.9                           | 22.0               |
| Drospirenone | Dog serum         | 1                    | 6.30                       | 59.6               | 105                          | 1.0                           | 100.0              |
|              | PON-AC            | 114                  | 0.14                       | 1.79               | 80.9                         | 0.7                           | 2.3                |

tein band was located at approximately 41 kDa referring to the three fraction pools (Fig. 4, gel 1, lanes B to D) as well as to native dog serum (Fig. 4, gel 1, lane A) as detected by Western blot. Some minor bands were found below the 41-kDa band (Fig. 4, gel 1, lane B and C). All the other PON-AC fractions and the negative control samples did not reveal any protein bands by Western blot (results not shown). The three groups of PON-AC fractions which were demonstrated to contain PON1 were pooled to determine their esteratic activity towards drospirenone and MPA (see 2.6). The activity yield concerning drospirenone cleavage was relatively low (2%) and the specific esterases were not enriched by PON-AC

(Table 5). In contrast, the isolation and purification of the MPAcleaving esterases by PON-AC were more successful as found by a relatively high activity yield (22%) and a purification factor of 7 (Table 5).

The SDS-PAGE gel of the MPA-cleaving albumin-AC fractions revealed some protein bands, indicating that the fractions were only partly purified (Fig. 4, gel 2, lane E). The Coomassie-stained protein bands of this gel were further investigated by protein identification using tryptic digestion and LC-ESI-MS technique. We were able to match the peptide mass fingerprint (obtained by MS) of the digested peptides of the 140-kDa band to canine  $\alpha_2$ -



**Fig. 4.** Gel 1: SDS-PAGE gel of native dog serum (lane A) and the three fraction pools (lanes B–D) from the PON-AC of dog serum. Lane A, B, C, and D contained 224, 51, 69, and 40  $\mu$ g protein, respectively. At approximately 41 kDa (see arrow), PON1 was detected in the lanes A to D by Western blot using an anti-PON1 antibody (see Section 2.13). Gel 2: SDS-PAGE gel of one active albumin-AC fraction from dog serum cleaving MPA (lane E). Lane E contained 1.1  $\mu$ g protein. The protein band at approximately 140 kDa (see arrow) was identified as  $\alpha_2$ -macroglobulin by tryptic digestion and subsequent LC-ESI-MS (see Section 2.12). The lane designated by "(St.)" shows the bands of the molecular-mass standard proteins as given in Fig. 3. Among them, myosin (191 kDa) was lost on gel handling concerning gel 1. In gel 1, the pre-stained standard proteins were visually detected. In gel 2, detection was accomplished by Coomassie Blue staining (see Section 2.10).

macroglobulin ( $\alpha_2$ M; NCBI no. gi|73997689). The protein score of 1941 for this match was significant (p < 0.05). The sequence coverage (i.e. the percentage of the theoretical amino acid sequence of the peptide found in the detected peptide sequence) amounted to approximately 20% while 38 peptides were matched.

# 4. Discussion

The aim of the present study was to study the stability of three structurally different and clinically relevant drug ester compounds in rat and dog serum and to find, where necessary, effective protection against the esteratic cleavage with the help of selected enzyme inhibitors. The investigated compounds were sagopilone (a macrocyclic lactone), drospirenone (a steroidal  $\gamma$ -butyrolactone), and MPA (a steroid with open-chain ester groups). To better understand the different efficiency of the tested inhibitors towards the drug-cleaving enzymes, we were interested in the identity of these enzymes. Therefore, we partly purified the esteratic fractions from serum of each species. As a first result, the isolation and purification of esteratic fractions from rat serum enabled us to identify ES-1 in the sagopilone-cleaving fractions. The purification of dog serum esterases led, furthermore, to the identification of  $\alpha_2 M$ in the fractions cleaving MPA. Likewise, PON1 was found in the drospirenone-cleaving fractions from dog serum. The found effective inhibitors can now be used for stabilization of these drugs in serum ex vivo.

### 4.1. Inhibition studies of native rat serum

Referring to the inhibition studies in native rat serum (Table 2), BNPP and Pefabloc<sup>®</sup> SC exhibited high efficacy for inhibition of sagopilone and MPA cleavage whereas drospirenone hydrolysis was less influenced. BNPP and Pefabloc® SC act as phosphorylating and sulfonylating agents, respectively, known to primarily inhibit serine hydrolases [18,46,47]. Therefore, we assume that serine hydrolases were mainly involved in the cleavage of sagopilone and MPA. Due to the lower, but still distinct capability of BNPP and Pefabloc<sup>®</sup> SC concerning drospirenone stabilization, this drug was probably not only cleaved by serine hydrolases but also by hydrolases without an active serine residue in their catalytic sides, such as PON. PON is a calcium-dependent hydrolase cleaving organophosphates (without being inhibited thereby) [48]. Moreover, sulfonyl fluorides are known to faintly inhibit PON [49]. The metal-ion chelating agent EDTA has been reported to strongly inhibit PON by complexation of calcium ions which are absolutely required for the activity and stability of PON [13,50]. In context with the relatively high efficiency of EDTA to stabilize drospirenone (as compared to the other drugs), we assume that PON was mainly responsible for the cleavage of drospirenone.

Although sagopilone- and MPA-cleaving enzymes were all equally sensitive towards serine hydrolase inhibitors, EDTA was able to discriminate between the drug-specific esterases. The lower efficiency of EDTA for sagopilone stabilization (as compared to MPA) let us assume that sagopilone was cleaved by CE whereas MPA was subjected to BChE activity. This was deduced from other studies reporting that EDTA was a more efficient inhibitor of BChE than of CE [51,52].

Zinc sulfate was partly efficient for the stabilization of MPA. The inhibition of BChE and some serine proteases by zinc ions has been reported [53–57]. Therefore, we assume that these hydrolases were mainly involved in MPA cleavage. However, it is also known that BChE inhibited by zinc ions can be partly reactivated in the presence of calcium ions (which were included in our pre-incubation buffer) [55,58]. Having this in mind, it could not be excluded that BChE was also responsible for sagopilone and drospirenone cleav-

age even though the stabilizing effect of zinc ions on these drugs was minor in our study.

TLCK is a histidine-modifying agent which generally forms covalent bonds with serine residues in serine hydrolases leading to irreversible inhibition [46]. In contrast to CE, it has been reported that BChE was only reversibly inhibited by TLCK [19,51,59,60]. As TLCK had virtually no effect on drospirenone or MPA cleavage, we assume that BChE was involved in these degradations. Sagopilone hydrolysis was markedly inhibited by TLCK so that BChE was probably not responsable for this cleavage.

Summarizing the results from the inhibition studies, we assume that sagopilone was hydrolyzed by CE whereas BChE was responsible for drospirenone and MPA cleavage. Moreover, drospirenone was cleaved by PON while serine proteases were involved in MPA cleavage.

#### 4.2. Inhibition studies of native dog serum

We found that sagopilone was not degraded in dog serum within the applied incubation time (see Section 2.6). This is in contrast to the results from rat serum where sagopilone was rapidly cleaved. We thus conclude that dog serum does not contain the CE that we assumed to be responsible for sagopilone cleavage in rat serum. This result is in accordance with others who reported a much greater CE activity as well as a high CE concentration in rat serum compared to dog serum as revealed by incubation and gel electrophoresis studies [15,16,27,30,61].

To effectively prevent MPA and drospirenone from enzymatic cleavage, selected enzyme inhibitors were studied in native dog serum. The serine hydrolase inhibitors Pefabloc<sup>®</sup> SC, APMSF, and BNPP exhibited marked inhibition of MPA degradation in dog serum (Table 3), thus pointing to the involvement of serine esterases in MPA cleavage.

Concerning drospirenone, there are good reasons to conclude that PON1 was involved in drospirenone hydrolysis as shown by the inhibition studies (Table 3). EDTA and EGTA are both known metal-ion chelating agents capable of mainly inhibiting metal-ion dependent enzymes (metallohydrolases) such as PON1.

The serine hydrolase inhibitors APMSF, BNPP, and Pefabloc<sup>®</sup> SC were less efficient than EDTA or EGTA, precluding any relevant involvement of serine hydrolases in drospirenone cleavage [46].

The high potency of NaF may be explained by precipitation of calcium fluoride due to the presence of calcium ions (in the buffer solution we used), leading to a markedly reduced calcium concentration and, subsequently, PON1 inhibition.

TBA was highly efficient in stabilizing drospirenone. As a possible reason, we assume that TBA (as a surface-active agent [62]) affected PON1 in such a way that this interaction resulted in conformational changes and inhibition of the enzyme as described for other detergents [63–65].

The great inhibitory effect of zinc and copper(II) ions on drospirenone cleavage also points to a major role of PON1 in this context [50,66]. This is supported by the finding of our study in rat serum where drospirenone cleavage was partly ascribed to rat serum PON1 (see Section 4.1). Furthermore, the structurally related spironolactone was also reported to be cleaved by PON isoenzymes [67,68].

# 4.3. Purification of esteratic fractions from rat serum and protein identification

Aiming to confirm our first suggestions on the identity of the drug-specific esterases and to understand the different inhibitor sensitivity, we partly purified the active fractions from rat serum to identify the responsible enzymes.

The applied three-step purification protocol to isolate and purify esterases from rat serum proved successful as shown by the high final purification factors (Table 4). Using an IEC as the first purification step was mainly based on the method of Tsuji et al. [19]. The direct application of serum to an anion-exchange column was additionally described by others [69,70]. Some reports on the isolation of esterases from rat serum or plasma employed fractionated precipitation with ammonium sulfate, followed by IEC [71-73]. We excluded the precipitation as this working step would mean a further physical burden on the proteins due to ultrafiltration or dialysis prior to IEC. The subsequent gel filtration (SEC) enabled us to estimate the molecular size of the drug-cleaving esterases to approximately 70 kDa or less (chromatogram not shown). By means of a dye-ligand affinity chromatography for removal of albumin (albumin-AC) using a Cibacron Blue F3G-A-coupled matrix, we obtained fractions greatly enriched in esteratic activities concerning the cleavage of the three drugs. The main esteratic activities were recovered in the flow-through fractions from this affinity chromatography (chromatogram not shown). This was in complete accordance with other reports on esterase isolation showing that BChE and CE did not bind to Cibacron Blue [30,74]. Due to the overall success of the purification sequence and the found activity distributions in the albumin-AC, we hypothesize that esterases, presumably BChE and/or CE, were involved in the cleavage of the drugs. Moreover, even serine proteases could be present in the active fractions owing to their ability to form tight complexes with BChE, being stable throughout isolation [75,76]. The low activity yields but high purification factors that we found were also experienced by others using an IEC as one of the first purification steps applied to esterases [19,30,71]. SDS-PAGE analyses of the active albumin-AC fractions revealed several protein bands (Fig. 3). The purification procedure could probably be improved to achieve fractions with higher purity than we did. For instance, affinity chromatography using specific antibodies or inhibitor ligands as well as immunoprecipitation may be useful as shown in other purification schemes [69,72,74,77,78].

As a first approach to protein identification, we identified the proteins of the SDS-PAGE gel of the albumin-AC fractions cleaving sagopilone. By tryptic digestion and MALDI-TOF-MS of the protein bands, we found the rat CE isoenzyme ES-1 contaminated with serum albumin in the band at approximately 70 kDa (Fig. 3 A). Both proteins were probably also present in the corresponding 68-kDa or 70-kDa protein band in the SDS-PAGE gels of the albumin-AC fractions cleaving drospirenone or MPA, respectively (Fig. 3 B and C). ES-1 is a member of the CES1C isoenzyme family [79]. CES1 isoenzymes are the main CE species in rat serum [72,79]. However, an isoenzyme belonging to the CES2 family has been detected in rat plasma, but only in a minor concentration [80]. Serum albumin that we identified either is known for its esterase-like activity [7,9,14]. However, it was not involved in the cleavage of the three drugs studied here since the albumin-containing fractions from albumin-AC eluted by the high-salt buffer did not exhibit relevant drug-cleaving activities (data not shown). From these results, we postulate that rat ES-1 could principally be involved in the cleavage of sagopilone (and probably both other drugs). Consistent with other reports on rat serum CE [30,72,74,77], we found a molecular mass of rat ES-1 of approximately 70 kDa by SEC (chromatogram not shown) as well as SDS-PAGE (Fig. 3A). It should be noted that we have yet only identified the *main* proteins in the bands of the SDS-PAGE gels. Therefore, our findings did not exclude that other esterases, especially BChE and serine proteases, were also present in the active fractions.

In all, our first hypotheses from the inhibition studies of native serum were supported by the results from protein isolation and identification. Summarizing the results from the protein purification, we conclude that the presence of serine hydrolases (i.e. CE, BChE, serine proteases) in the active fractions was likely. In fact, ES-1 was subsequently identified in the esteratic fractions cleaving sagopilone. Although the identification of drospirenoneand MPA-cleaving esterases and further studies on the identity of sagopilone-cleaving esterases remain to be carried out, we assume that sagopilone was cleaved by CE, drospirenone by BChE and PON while MPA was degraded by BChE and serine proteases. These assumptions would also explain the inhibitor sensitivity since, for instance, the assumed enzyme for sagopilone cleavage (i.e. ES-1) would be greatly inhibited by serine hydrolase inhibitors (BNPP, Pefabloc<sup>®</sup> SC) as we found here indeed.

Regarding drospirenone, it has been reported that the structurally related drug spironolactone was cleaved by human PON1 and PON3, thus supporting our suggestion of the involvement of PON in drospirenone cleavage [67,68]. As PON3 has solely been found in rat hepatic microsomes yet, we assume that rat PON1 was responsible for this hydrolysis in serum [68,81]. With regard to structural MPA analogues, the charged steroid-21-monoester methylprednisolone hemisuccinate was mainly cleaved by a rat CES2 isoenzyme [80], thus assuming different substrate specificities of rat BChE and CE isoenzymes. Indeed, the uncharged 21-monoester methylprednisolone acetate, but not the hemisuccinate ester, was described to be hydrolyzed by human BChE [82] which supports our assumption of the involvement of BChE in MPA cleavage. To our best knowledge, studies of enzymes cleaving sagopilone or structurally related compounds have not yet been published.

# 4.4. Purification of esteratic fractions from dog serum and protein identification

For isolation and identification of the MPA-specific esterases from dog serum, we applied a three-step purification scheme as outlined before (see Section 4.3). The purification cascade was successful concerning MPA-cleaving esterases as shown by the high final purification factor (Table 5). We found  $\alpha_2$  M at 140 kDa (Fig. 4, gel 2, lane E) by protein identification using LC-ESI-MS of the SDS-PAGE gel of one active albumin-AC fraction. The endogenous protease inhibitor  $\alpha_2$ M has no reported *intrinsic* esterase activity. We therefore assume that  $\alpha_2 M$  formed a complex with a serine protease or serine esterase, but retaining the capability of the bound protease or esterase to cleave MPA [83,84]. The protease inhibitor  $\alpha_2$ M is a homotetramer with a molecular mass of 180 kDa for each monomer which can be fragmentated into smaller species upon sample preparation for SDS-PAGE [84,85]. We thereby explain the found molecular mass of 140 kDa in the present study. This complex seemed to be stable throughout the purification steps since the main MPA-cleaving activity eluted in the SEC in the range of high molecular mass proteins (>484 kDa). Moreover, it is known that  $\alpha_2$ M strongly binds to Cibacron Blue-coupled matrices, thereby explaining the activity distribution of the albumin-AC fractions [86,87].

With respect to the identification of MPA-specific serine esterase, we hypothesize that BChE was involved in this cleavage. This is mainly based on our results from rat serum (see Section 4.3) as well as on the finding that the structural analogue methylprednisolone-21-acetate was reported to be cleaved by this enzyme [82]. In addition, our hypothesis is in full accordance with the high efficiency of the serine hydrolase inhibitors Pefabloc<sup>®</sup> SC, APMSF, and BNPP. It was, moreover, reported that BChE can complex with  $\alpha_2$ M while retaining its esterase activity even on native PAGE gels [7]. BChE can, however, not alone be responsible for MPA cleavage because, in this case, one would have seen much greater inhibition by NaF and lower inhibition by TLCK in the presence of specific substrates than we actually did [59,88]. It has been described that BChE possesses two functionally different enzymatic sites; one is responsible for the BChE esterase activity while the other is for the aryl acylamidase activity [75,89,90]. Both sites are overlapping but not identical as found by studies of the influence of certain enzyme inhibitors and activators on these functions [69,90,91]. The two enzymatic activities are, however, inhibited by serine hydrolase inhibitors [30,69]. Although the effects of NaF and TLCK on the amidase function remain to be investigated, we assume that the amidatic site of BChE was responsible for MPA cleavage. This hypothesis is based on the fact that the found influence of both inhibitors was in contrast to what would have been expected for their influence on the esterase function of BChE. MPA is, of course, not an amide but both enzymatic sites of BChE are in close contact and possess a catalytically active serine residue, thus probably enabling the amidatic site to readily cleave esters such as MPA [90,92].

Further evidence for the involvement of PON1 in drospirenone cleavage in dog serum came from the purification methods we applied, thus supporting the results from the inhibition studies. Being successful for isolation of MPA-cleaving enzymes, the three-step purification procedure, however, resulted in a complete activity loss after the SEC step concerning the enzymes hydrolyzing drospirenone. Due to our hypothesis of the involvement of PON1, we used a relatively specific method for isolation of PON1 using Cibacron Blue affinity chromatography (PON-AC) which was basically described by Gan et al. [40]. We obtained several fractions being part of peaks in the UV chromatogram (not shown). These fractions were investigated by SDS-PAGE and, subsequently, Western blotting using an anti-PON1 antibody. In contrast to Gan et al. [40], we found by Western blot that PON1 already eluted from the affinity column due to washing with a NaCl-free buffer. The remaining esterase was then desorbed from the column by elution with sodium deoxycholate and restoration of the initial conditions. The early desorption of PON1 from the column was probably due to the lower NaCl concentration that we used as compared to that of Gan et al. (2 M vs. 3 M NaCl) [40]. Therefore, PON1 might not have been firmly adsorbed to the matrix and, as a consequence, it was more readily desorbed. We had to substitute the buffer containing 3 M NaCl (used by Gan et al. [40]) by a buffer containing 2 M NaCl since our purification system did not tolerate the high viscosity of the former buffer. We detected PON1 in the SDS-PAGE gel of native dog serum as well as of the mentioned PON-AC fractions at approximately 41 kDa (Fig. 4, gel 1, lanes A-D) which fully corresponds to other reported molecular masses of serum PON (laying in the range of 37-48 kDa). [93-95]. The presence of some minor PON1 bands below the main 41-kDa band in lane B and C (Fig. 4) could be explained by the fact that PON1 may exist in different glycosylated isoforms [93,94]. To our best knowledge, we here report for the first time the detection of PON1 in dog serum although its presence has long been suggested owing to incubation and gel electrophoresis studies with PON substrates [27,29,96,97]. The amino acid sequence of dog PON1 has been deduced from the corresponding complementary deoxyribonucleic acid (cDNA) sequence [98]. The primary antibody that we used for Western blot (see Section 2.13) was raised against a synthetic peptide of 12 amino acids representing a part of the sequence of human PON1. It is known that anti-PON antibodies exhibit cross-reactivity towards other PON isoenzymes within a given species and even between different species due to the high intra- and interspecific sequence homology of PON isoenzymes (reaching 60-86%) [13,98-100], clearly explaining the positive (cross-)reaction towards dog serum PON1 in our study.

We used the three mentioned groups of PON1-containing fractions from the PON-AC (see Section 3.4) as one pool to determine their esteratic activity towards drospirenone or MPA. However, the activity yield and purification factor of these fractions concerning drospirenone cleavage were low as compared to MPA (Table 5). One reason for this finding may be the fact that the cleaved lactone ring of drospirenone may tend to re-cyclize in aqueous solutions as described for other  $\gamma$ -butyrolactones [101,102]. PON1 generally exhibits lower specific activities towards lactones compared with aromatic non-cyclic esters [67,103]. In addition, dog serum was reported to have lower catalytic efficiencies towards aromatic esters in comparison to rat serum, for instance [29,104]. All these factors might have resulted in a true and/or artificial low activity of the PON1-containing fractions in terms of drospirenone cleavage.

The PON1-containing fractions from PON-AC gave a high activity yield and considerable enrichment of esterases cleaving MPA as compared to drospirenone (Table 5). Although we have demonstrated that PON1 was present in these fractions (see above), PON1 was not assumed to be responsible for MPA cleavage. This hypothesis is mainly based on the finding that PON1 was able to attack estradiol-3,17 $\beta$ -diesters cleaving the ester group at position 3



**Fig. 5.** Efficiency of selected enzyme inhibitors to prevent sagopilone, drospirenone, and MPA from esteratic degradation in native rat (A) or dog serum (B). Data were selected from Tables 2 and 3.

but not that one at position  $17\beta$  (unless there was an adjacent enolization in ring D of the steroid structure) [105]. Moreover, methylprednisolone-21-acetate, a structural analogue to MPA, was found not to be cleaved by human PON1 [82]. Further reasons to conclude that PON1 did not cleave MPA can be brought about by our results from the inhibition studies (Table 3): The metallohydrolase inhibitors EDTA and EGTA had no relevant inhibitory effect on MPA cleavage. The high specific activity of the PON1-containing fractions towards MPA can probably be explained by the fact that other esterases may be isolated along with PON1 using Cibacron Blue affinity chromatography as reported, for instance, for platelet-activating factor acetylhydrolase (PAF-AH) [106,107].

In all, we assume that drospirenone was cleaved by dog serum PON1 whereas MPA was degraded by BChE.

# 5. Conclusion

To summarize, we successfully found efficient enzyme inhibitors for the stabilization of sagopilone, drospirenone, and MPA in rat serum as well as of drospirenone and MPA in dog serum. Moreover, the inhibitor sensitivity of the drug-specific esterases in native serum, in conjunction with the (first) results from protein purification and identification, shed some light on the identity of the cleaving esterases, thereby explaining the different efficiency of the tested inhibitors. Further studies are clearly needed to have sound evidence on the actual involvement of the assumed enzymes in drug cleavage.

Our results from the inhibition experiments (Tables 2 and 3, Fig. 5) can be used to stabilize these and structurally related compounds in rat and canine serum, and probably blood, samples *ex vivo* during preclinical studies. As a straightforward approach, one should extend the use of the found inhibitors to improve the stability of chemically unrelated drug esters in blood of other test species as well as in human blood for exploring the efficiency of the inhibitors in preclinical and clinical studies with newly developed drug ester compounds.

# **Conflict of interest**

Matthias Koitka was a former research fellow at Bayer Schering Pharma AG, Berlin, Germany. Joachim Höchel and Hille Gieschen are full-time employees at Bayer Schering Pharma AG, Berlin, Germany. Hans-Hubert Borchert has no conflicts of interest to disclose.

# Acknowledgements

We greatly acknowledge the generous support of Mr. Norbert Otto from the Department of Protein Supply at Bayer Schering Pharma AG, Berlin, in terms of protein purification and analysis. Regarding protein identification, we thank Dr. Inke Bahr from the Department of Target Research for her excellent work. Mrs. Birgit Albrecht and Dr. Karsten Denner from the Department of Drug Metabolism are gratefully acknowledged for the opportunity to use their equipment for protein analysis. Finally, we appreciate the helpful assistance of Mrs. Sabine Backhus from the Department of Nonclinical Pharmacokinetics, Berlin, in radiochemical HPLC analysis.

#### References

- U.S. Department of Health and Human Services, Food and Drug Administration, Guidance for industry, Bioanalytical method validation, 2001.
- [2] The European Agency for the Evaluation of Medicinal Products, Pharmacokinetic studies in man, Guideline 3CC3a, 1987. Internet: http://www. emea.europa.eu/htms/human/humanguidelines/efficacy.htm (as of June 23rd, 2009).

- [3] The European Agency for the Evaluation of Medicinal Products, Conduct of pharmacokinetic studies in animals, Guideline 7AE3a, 1992. Internet: http://ec.europa.eu/enterprise/pharmaceuticals/eudralex/vol7\_en.htm#7a (as of June 23rd, 2009).
- [4] The European Agency for the Evaluation of Medicinal Products, Committee for Veterinary Medicinal Products, Guidelines for the conduct of pharmacokinetic studies in target animal species, EMEA/CVMP/133/99, London, 2000.
- [5] T. Tokumura, A. Muraoka, T. Masutomi, Y. Machida, Stability of spironolactone in rat plasma: strict temperature control of blood an plasma samples is required in rat pharmacokinetic studies, Biol. Pharm. Bull. 28 (2005) 1126–1128.
- [6] J. Zeng, D. Onthank, P. Crane, S. Unger, N. Zheng, S. Pasas-Farmer, M. Arnold, Simultaneous determination of a selective adenosine 2A agonist, BMS-068645, and its acid metabolite in human plasma by liquid chromatography-tandem mass spectrometry—evaluation of the esterase inhibitor, diisopropyl fluorophosphate, in the stabilization of a labile ester-containing drug, J. Chromatogr. B 852 (2007) 77–84.
- [7] B. Li, M. Sedlaček, I. Manoharan, R. Boopathy, E.G. Duysen, P. Masson, O. Lockridge, Butyrylcholinesterase, paraoxonase, and albumin esterase, but not carboxylesterase, are present in human plasma, Biochem. Pharmacol. 70 (2005) 1673–1684.
- [8] F.M. Williams, Serum enzymes of drug metabolism, Pharmacol. Ther. 34 (1987) 99–109.
- [9] B. Testa, J.M. Mayer, Hydrolysis in Drug and Prodrug Metabolism. Chemistry, Biochemistry, and Enzymology, Verlag Helvetica Chemica Acta, Zürich, 2003.
- [10] Nomenclature Committee of the International Union of Biochemistry and Molecular Biology (IUBMB), See http://www.chem. qmul.ac.uk/iubmb/enzyme for continuously updated versions.
- [11] S. Brimijoin, P. Hammond, Butyrylcholinesterase in human brain and acetylcholinesterase in human plasma: trace enzymes measured by two-site immunoassay, J. Neurochem. 51 (1988) 1227–1231.
- [12] A.T. Boeck, L.M. Schopfer, O. Lockridge, DNA sequence of butyrylcholinesterase from the rat: expression of the protein and characterization of the properties of rat butyrylcholinesterase, Biochem. Pharmacol. 63 (2002) 2101–2110.
- [13] M. Harel, A. Aharoni, L. Gaidukov, B. Brumshtein, O. Khersonsky, R. Meged, H. Dvir, R.B.G. Ravelli, A. McCarthy, L. Toker, I. Silman, J.L. Sussman, D.S. Tawfik, Structure and evolution of the serum paraoxonase family of detoxifying and anti-atherosclerotic enzymes, Nat. Struct. Mol. Biol. 11 (2004) 412– 419.
- [14] B.M. Liederer, R.T. Borchardt, Enzymes involved in the bioconversion of esterbased prodrugs, J. Pharm. Sci. 95 (2006) 1177–1195.
- [15] S.H. Lawrence, P.J. Melnick, H.E. Weimer, A species comparison of serum proteins and enzymes by starch gel electrophoresis, Proc. Soc. Exp. Biol. Med. 105 (1960) 572–575.
- [16] K. Augustinsson, Electrophoresis studies on blood plasma esterases. I: Mammalian plasmata, Acta Chem. Scand. 13 (1959) 571–592.
- [17] B.M. Liederer, R.T. Borchardt, Stability of oxymethyl-modified coumarinic acid cyclic prodrugs of diastereomeric opioid peptides in biological media from various animal species including human, J. Pharm. Sci. 94 (2005) 2198–2206.
- [18] T. Minagawa, Y. Kohno, T. Suwa, A. Tsuji, Species differences in hydrolysis of isocarbacyclin methyl ester (TEI-9090) by blood esterases, Biochem. Pharmacol. 49 (1995) 1361–1365.
- [19] T. Tsuji, N. Kaneda, K. Kado, T. Yokokura, T. Yoshimoto, D. Tsuru, CPT-11 converting enzyme from rat serum: purification and some properties, J. Pharmacobiodyn. 14 (1991) 341–349.
- [20] P.D. Senter, H. Marquardt, B.A. Thomas, B.D. Hammock, I.S. Frank, H.P. Svensson, The role of rat serum carboxylesterase in the activation of paclitaxel and camptothecin prodrugs, Cancer Res. 56 (1996) 1471–1474.
- [21] J.A. Crow, A. Borazjani, P.M. Potter, M.K. Ross, Hydrolysis of pyrethroids by human and rat tissues: examination of intestinal, liver and serum carboxylesterases, Toxicol. Appl. Pharmacol. 221 (2007) 1–12.
- [22] C.Y. Quon, H.F. Stampfli, Biochemical properties of blood esmolol esterase, Drug Metab. Dispos. 13 (1985) 420–424.
- [23] C.Y. Quon, K. Mai, G. Patil, H.F. Stampfli, Species differences in the stereoselective hydrolysis of esmolol by blood esterases, Drug Metab. Dispos. 16 (1988) 425–428.
- [24] C.Y. Quon, H.F. Stampfli, Biochemical characterization of flestolol esterase, Res. Commun. Chem. Pathol. Pharmacol. 81 (1993) 309–322.
- [25] M.M. Reidenberg, The procaine esterase activity of serum from different mammalian species, Proc. Soc. Exp. Biol. Med. 140 (1972) 1059–1061.
- [26] Y. Okuyama, K. Momota, A. Morino, Pharmacokinetics of prulifloxacin. 1st communication: absorption, distribution and excretion in rats, dogs and monkeys after a single administration, Drug Res. 47 (1997) 276–284.
- [27] K. Augustinsson, Electrophoretic separation and classification of blood plasma esterases, Nature 181 (1958) 1786–1789.
- [28] K. Augustinsson, Multiple forms of esterase in vertebrate blood plasma, Ann. N. Y. Acad. Sci. 94 (1961) 844–860.
- [29] D. Ecobichon, Characterization of the esterases of canine serum, Can. J. Biochem. 48 (1970) 1359–1367.
- [30] T. Tsujita, H. Okuda, Carboxylesterases in rat and human sera and their relationship to serum aryl acylamidases and cholinesterases, Eur. J. Biochem. 133 (1983) 215–220.
- [31] D. Vincent, H. Perrier, J.M. Rouzioux, Sur les isoenzymes de la cholinestérase sérique chez quelques espèces animales, C.R. Séances Soc. Biol. Fil. 164 (1970) 1767–1769 (in French).

- [32] W.E. Hoffmann, P.F. Solter, B.W. Wilson, Clinical enzymology, in: W.F. Loeb, F.W. Quimby (Eds.), The Clinical Chemistry of Laboratory Animals, second ed., Taylor & Francis, Philadelphia, 1999, pp. 399–454.
- [33] M.R. Redinbo, P.M. Potter, Mammalian carboxylesterases: from drug targets to protein therapeutics, Drug Discov. Today 10 (2005) 313–325.
- [34] N. Lindegardh, G.R. Davies, T.T. Hien, J. Farrar, P. Singhasivanon, N.P.J. Day, N.J. White, Importance of collection tube during clinical studies of oseltamivir, Antimicrob. Agents Chemother. 51 (2007) 1835–1836.
- [35] U. Klar, B. Buchmann, W. Schwede, W. Skuballa, J. Hoffmann, R.B. Lichtner, Total synthesis and antitumor activity of ZK-EPO: the first fully synthetic epothilone in clinical development, Angew. Chem. Int. Ed. 45 (2006) 7942–7948.
- [36] R. Krattenmacher, Drospirenone: pharmacology and pharmacokinetics of a unique pregestogen, Contraception 62 (2000) 29–38.
- [37] T. Ruzicka, Methylprednisolone aceponate in eczema and other inflammatory skin disorders - a clinical update, Int. J. Clin. Pract. 60 (2006) 85–92.
- [38] M. Koitka, J. Höchel, D. Obst, A. Rottmann, H. Gieschen, H.H. Borchert, Determination of rat serum esterase activities by an HPLC method using S-acetylthiocholine iodide and p-nitrophenyl acetate, Anal. Biochem. 381 (2008) 113–122.
- [39] R.K. Scopes, Protein Purification. Principles and Practice, third ed., Springer, New York, 1994.
- [40] K.N. Gan, A. Smolen, H.W. Eckerson, B.N. La Du, Purification of human serum paraoxonase/arylesterase, Drug Metab. Dispos. 19 (1991) 100–106.
- [41] P.K. Smith, R.I. Krohn, G.T. Hermanson, A.K. Mallia, F.H. Gartner, M.D. Provenzano, E.K. Fujimoto, N.M. Goeke, B.J. Olson, D.C. Klenk, Measurement of protein using bicinchoninic acid, Anal. Biochem. 150 (1985) 76–85.
- [42] Pierce Biotechnology, Micro BCA Protein Assay Kit, Instructions, Rockford, 2006.
- [43] S.R. Gallagher, One-dimensional electrophoresis using nondenaturing conditions, Current Protocols in Protein Science, Unit 10.3, 1995, doi:10.1002/0471140864.ps1003s00.
- [44] Matrix Science, Internet: http://www.matrixscience.com/cgi/master\_results. pl?file=.%2Fdata%2FF981122.dat&REP TYPE=concise&\_sigthreshold=0.05& REPORT=AUTO (as of June 21st, 2009).
- [45] Invitrogen, iBlot Dry Blotting System, User Manual, Carlsbad, 2007.
- [46] J.C. Powers, J.W. Harper, Inhibitors of serine proteinases, in: A.J. Barrett, G. Salvesen (Eds.), Proteinase Inhibitors, Elsevier, Amsterdam, 1986, pp. 55–152.
- [47] G.R. Mintz, An irreversible serine protease inhibitor, BioPharm 6 (1993) 34–38.
- [48] A.L. Pond, C.P. Coyne, H.W. Chambers, J.E. Chambers, Identification and isolation of two rat serum proteins with A-esterase activity toward paraoxon and chlorpyrifos-oxon, Biochem. Pharmacol. 52 (1996) 363–369.
- [49] C. Jewell, C. Ackermann, N.A. Payne, G. Fate, R. Voorman, F.M. Williams, Specificity of procaine and ester hydrolysis by human, minipig, and rat skin and liver, Drug Metab. Dispos. 35 (2007) 2015–2022.
- [50] C.L. Kuo, B.N. La Du, Calcium binding by human and rabbit serum paraoxonases. Structural stability and enzymatic activity, Drug Metab. Dispos. 26 (1998) 653–660.
- [51] Y. Fujita, T. Yaegashi, S. Sawada, H. Oyama, T. Yoshimoto, D. Tsuru, Partial purification and characterization of an esterase acting on the anticancer prodrugs, 7-ethylcamptothecin derivatives, Biol. Pharm. Bull. 18 (1995) 648– 652.
- [52] C.D. Bhanumathy, A.S. Balasubramanian, Evidence for a Zn<sup>2+</sup>-binding site in human serum butyrylcholinesterase, Biochem. J. 315 (1996) 127–131.
- [53] H. Zollner, Handbook of Enzyme Inhibitors, third ed., Wiley-VCH, Weinheim, 1999.
- [54] A.J. Barrett, Introduction to the history and classification of tissue proteinases, in: A.J. Barrett (Ed.), Proteinases in Mammalian Cells and Tissues, Elsevier/North-Holland Biomedical Press, Amsterdam, 1977, pp. 1–55.
- [55] B. Sarkarati, A.N. Cokugras, E.F. Tezcan, Inhibition kinetics of human serum butyrylcholinesterase by Cd<sup>2+</sup>, Zn<sup>2+</sup> and Al<sup>3+</sup>: comparison of the effects of metal ions on cholinesterases, Comp. Biochem. Physiol. Part C 122 (1999) 181–190.
- [56] C.A. Borgono, I.P. Michael, J.L.V. Shaw, L.Y. Luo, M.C. Gosh, A. Soosaipillai, L. Grass, D. Katsaros, E.P. Diamandis, Expression and functional characterization of the cancer-related serine protease, human tissue kallikrein 14, J. Biol. Chem. 282 (2007) 2405–2422.
- [57] L.C. Petersen, O.H. Olsen, L.S. Nielsen, P.O. Freskgard, E. Persson, Binding of Zn<sup>2+</sup> to a Ca<sup>2+</sup> loop allosterically attenuates the activity of factor VIIa and reduces its affinity for tissue factor, Protein Sci. 9 (2000) 859–866.
- [58] A.N. Cokugras, E.F. Tezcan, Inhibition kinetics of brain butyrylcholinesterase by Cd<sup>2+</sup> and Zn<sup>2+</sup>, Ca<sup>2+</sup> or Mg<sup>2+</sup> reactivates the inhibited enzyme, Int. J. Biochem. 25 (1993) 1115–1120.
- [59] D. Cengiz, A.N. Cokugras, K. Kilinc, E.F. Tezcan, Histidine modification of human serum butyrylcholinesterase, Biochem. Mol. Med. 61 (1997) 52–57.
- [60] O.T. Oboh, N.S. Lamago, Liver prenylated methylated protein methyl esterase is the same enzyme as Sus scrofa carboxylesterase, J. Biochem. Mol. Toxicol. 22 (2008) 51–62.
- [61] D. Richter, P.G. Croft, Blood esterases, Biochem. J. 36 (1942) 746-757.
- [62] P. Mukerjee, K.J. Mysels, Critical micelle concentration of aqueous surfactant systems, Nat. Stand. Ref. Data Ser., Nat. Bur. Stand. (U.S.) 36 (1971), Washington.
- [63] O. Khersonsky, D.S. Tawfik, Structure-reactivity studies of serum paraoxonase PON1 suggest that its native activity is lactonase, Biochemistry 44 (2005) 6371–6382.

- [64] C.E. Furlong, R.E. Richter, C. Chapline, J.W. Crabb, Purification of rabbit and human serum paraoxonase, Biochemistry 30 (1991) 10133–10140.
- [65] Y.S. Huang, L. Woods, L.G. Sultatos, Solubilization and purification of Aesterase from mouse hepatic microsomes, Biochem. Pharmacol. 48 (1994) 1273–1280.
- [66] F. Gil, M.C. Gonzalvo, A.F. Hernandez, E. Villanueva, A. Pla, Differences in the kinetic properties, effect of calcium and sensitivity to inhibitors of paraoxon hydrolase activity in rat plasma and microsomal fraction from rat liver, Biochem. Pharmacol. 48 (1994) 1559–1568.
- [67] S. Billecke, D. Draganov, R. Counsell, P. Stetson, C. Watson, C. Hsu, B.N. La Du, Human serum paraoxonase (PON1) isozymes Q and R hydrolyze lactones and cyclic carbonate esters, Drug Metab. Dispos. 28 (2000) 1335–1342.
- [68] D.I. Draganov, J.F. Teiber, A. Speelman, Y. Osawa, R. Sunahara, B.N. La Du, Human paraoxonases (PON1, PON2, and PON3) are lactonases with overlapping and distinct substrate specificities, J. Lipid Res. 46 (2005) 1239– 1247.
- [69] S.T. George, A.S. Balasubramanian, The aryl acylamidases and their relationship to cholinesterases in human serum, erythrocyte and liver, Eur. J. Biochem. 121 (1981) 177–186.
- [70] R.V. Rao, A.S. Balasubramanian, Localization of the peptidase activity of human serum butyrylcholinesterase in a -50-kDa fragment obtained by limited α-chymotrypsin digestion, Eur. J. Biochem. 188 (1990) 637–643.
- [71] M. Hashinotsume, K. Higashino, T. Hada, Y. Yamamura, Purification and enzymatic properties of rat serum carboxylesterase, J. Biochem. 84 (1978) 1325–1333.
- [72] B. Yan, D. Yang, P. Bullock, A. Parkinson, Rat serum carboxylesterase. Cloning, expression, regulation, and evidence of secretion from rat liver, J. Biol. Chem. 270 (1995) 19128–19134.
- [73] H.A. Van Lith, M. Haller, I.J.M. Van Hoof, M.J.A. Van Der Wouw, B.F.M. Van Zutphen, A.C. Beynen, Characterization of rat plasma esterase ES-1A concerning its molecular and catalytic properties, Arch. Biochem. Biophys. 301 (1993) 265–274.
- [74] T. Hirose, A. Morita, M. Yamada, H. Nikaido, J. Hayakawa, O. Nikaido, Selective purification of sex-influenced esterase from rat serum by immunoaffinity chromatographies, Eur. J. Biochem. 189 (1990) 431–435.
- [75] S. Darvesh, R. Kumar, S. Roberts, R. Walsh, E. Martin, Butyrylcholinesterasemediated enhancement of the enzymatic activity of trypsin, Cell. Mol. Neurobiol. 21 (2001) 285–296.
- [76] F. Checler, J. Grassi, J.P. Vincent, Cholinesterases display genuine arylacylamidase activity but are totally devoid of intrinsic peptidase activities, J. Neurochem. 62 (1994) 756–763.
- [77] S.E.H. Alexson, T.H. Finlay, U. Hellman, L.T. Svensson, U. Diczfalusy, G. Eggertsen, Molecular cloning and identification of a rat carboxylesterase expressed in the liver, J. Biol. Chem. 269 (1994) 17118–17124.
- [78] T. Tsujita, K. Nagai, H. Okuda, Purification and properties of human serum esterase, Biochim. Biophys. Acta 570 (1979) 88–95.
- [79] T. Satoh, M. Hosokawa, The mammalian carboxylesterases: from molecules to functions. Ann. Rev. Pharmacol. Toxicol. 38 (1998) 257-288.
- [80] T. Furihata, M. Hosokawa, Dexamethasone-induced methylprednisolone hemisuccinate hydrolase: its identification as a member of the rat carboxylesterase 2 family and its unique existence in plasma, Biochem. Pharmacol. 69 (2005) 1287–1297.
- [81] L. Rodrigo, F. Gil, A.F. Hernandez, O. Lopez, A. Pla, Identification of paraoxonase 3 in rat liver microsomes: purification and biochemical properties, Biochem. J. 376 (2003) 261–268.
- [82] C. Myers, O. Lockridge, B.N. La Du, Hydrolysis of methylprednisolone acetate by human serum cholinesterase, Drug Metab. Dispos. 10 (1982) 279–280.
- [83] G.S. Salvesen, H. Nagase, Inhibition of proteolytic enzymes, in: R. Beynon, J.S. Bond (Eds.), Proteolytic Enzymes, second ed., Oxford University Press, Oxford, 2001, pp. 105–130.
- [84] L. Sottrup-Jensen, α<sub>2</sub>-Macroglobulin and related thiol ester plasma proteins, in: F. Putnam (Ed.), The Plasma Proteins, vol. V, second ed., Academic Press, Orlando, 1987, pp. 191–291.
- [85] M. Andersson, U. Jönsson, A. Olsson, A slow form of alpha-2-macroglobulin in diseased and healthy dogs, J. Comp. Pathol. 127 (2002) 37–44.
- [86] J. Travis, R. Pannell, Selective removal of albumin from plasma by affinity chromatography, Clin. Chim. Acta 49 (1973) 49–52.
- [87] J. Naval, M. Čalvo, F. Lampreave, A. Pineiro, Interactions of different albumins and animal sera with insolubilized Cibacron Blue. Evaluation of apparent affinity constants, Comp. Biochem. Physiol. 71B (1982) 403–407.
- [88] G. Cimasoni, Inhibition of cholinesterases by fluoride in vitro, Biochem. J. 99 (1966) 133–137.
- [89] M.F. Montenegro, M.T. Moral-Naranjo, M. Páez de la Cadena, F.J. Campoy, E. Muñoz-Delgado, C.J. Vidal, The level of aryl acylamidase activity displayed by human butyrylcholinesterase depends on its molecular distribution, Chem. Biol. Interact. 175 (2008) 336–339.
- [90] R.V. Rao, A.S. Balasubramanian, Isolation of a galactose-free 20-kDa fragment exhibiting butyrylcholine esterase and aryl acylamidase activity from human serum butyrylcholine esterase by limited α-chymotrypsin digestion, Eur. J. Biochem. 179 (1989) 639–644.
- [91] R. Boopathy, A.S. Balasubramanian, Chemical modification of the bifunctional human serum pseudocholinesterase, Eur. J. Biochem. 151 (1985) 351–360.
- [92] S. Darvesh, R.S. McDonald, K.V. Darvesh, D. Mataija, S. Mothana, H. Cook, K.M. Carneiro, N. Richard, R. Walsh, E. Martin, On the active site for hydrolysis of aryl amides and choline esters by human cholinesterases, Bioorg. Med. Chem. 14 (2006) 4586–4599.

- [93] L. Golmanesh, H. Mehrani, M. Tabei, Simple procedures for purification and stabilization of human serum paraoxonase-1, J. Biochem. Biophys. Methods 70 (2008) 1037–1042.
- [94] C.L. Kuo, B.N. La Du, Comparison of purified human and rabbit paraoxonases, Drug Metab. Dispos. 23 (1995) 935–944.
- [95] D.I. Draganov, B.N. La Du, Pharmacogenetics of paraoxonases: a brief review, Naunyn-Schmiedeberg's Arch. Pharmacol. 369 (2004) 78-88.
- [96] S. Motta, C. Letellier, M. Ropert, C. Motta, J.J. Thiébault, Protecting effect of vitamin E supplementation on submaximal exercise-induced oxidative stress in sedentary dogs as assessed by erythrocyte membrane fluidity and paraoxonase-1 activity, Vet. J. 181 (2009) 288–295.
- [97] C.A. Leone, R.L. Anthony, Serum esterases among registered breeds of dogs as revealed by immunoelectrophoretic comparisons, Comp. Biochem. Physiol. 18 (1966) 359–368.
- [98] S.L. Primo-Parmo, R.C. Sorenson, J. Teiber, B.N. La Du, The human serum paraoxonase/arylesterase gene (PON1) is one member of a multigene family, Genomics 33 (1996) 498–507.
- [99] P.W. Connelly, G.F. Maguire, C.M. Picardo, J.F. Teiber, D. Draganov, Development of an immunoblot assay with infrared fluorescence to quantify paraoxonase 1 in serum and plasma, J. Lipid Res. 49 (2008) 245–250.
- [100] E. Thomas-Moya, M. Gianotti, I. Llado, A.M. Proenza, Effects of caloric restriction and gender on rat serum paraoxonase 1 activity, J. Nutr. Biochem. 17 (2006) 197–203.

- [101] G. Rousseau, Medium ring lactones, Tetrahedron 51 (1995) 2777-2849.
- [102] W.N. Fishbein, S.P. Bessman, Purification and properties of an enzyme in human blood and rat liver microsomes catalyzing the formation and hydrolysis of γ-lactones. II: Metal ion effects, kinetics, and equilibria, J. Biol. Chem. 241 (1966) 4842–4847.
- [103] O. Khersonsky, D.S. Tawfik, The histidine 115-histidine 134 dyad mediates the lactonase activity of mammalian serum paraoxonases, J. Biol. Chem. 281 (2006) 7649–7656.
- [104] N.W. McCracken, P.G. Blain, F.M. Williams, Nature and role of xenobiotic metabolizing esterases in rat liver, lung, skin and blood, Biochem. Pharmacol. 45 (1993) 31–36.
- [105] J.F. Teiber, S.S. Billecke, B.N. La Du, D.I. Draganov, Estrogen esters as substrates for human paraoxonases, Arch. Biochem. Biophys. 461 (2007) 24–29.
- [106] G.K. Marathe, G.A. Zimmerman, T.M. McIntyre, Platelet-activating factor acetylhydrolase, and not paraoxonase-1, is the oxidized phospholipid hydrolase of high density lipoprotein particles, J. Biol. Chem. 278 (2003) 3937– 3947.
- [107] T. Kriska, G.K. Marathe, J.C. Schmidt, T.M. McIntyre, A.W. Girotti, Phospholipase action of platelet-activating factor acetylhydrolase, but not paraoxonase-1, on long fatty acyl chain phospholipid hydroperoxides, J. Biol. Chem. 282 (2007) 100–108.